Author's response to reviews

Title: Magnetic resonance imaging, computed tomography, and 68Ga-DOTATOC positron emission tomography for imaging skull base meningiomas with infracranial extension treated with stereotactic radiotherapy - a case series

Authors:

Reinhold Graf (reinhold.graf@charite.de)
Michail Plotkin (michail.plotkin@charite.de)
Ingo G Steffen (ingo.steffen@charite.de)
Reinhard Wurm (reinhard.wurm@klinikumffo.de)
Peter Wust (peter.wust@charite.de)
Winfried Brenner (winfried.brenner@charite.de)
Volker Budach (volker.budach@charite.de)
Harun Badakhshi (harun.badakhshi@charite.de)

Version: 2 Date: 31 December 2011

Author's response to reviews:

Abstract: This should be identical in the manuscript and the submission system. Currently the conclusion has not been added to the submission system abstract.

Ethical approval: Any experimental research that is reported in the manuscript should have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration. A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.

Consent: Please include a consent statement including the following information: `Written informed consent was obtained from the patient for publication of this case report and accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal.